Author's response to reviews

Title: Recommendations to improve physical activity among teenagers- A qualitative study with ethnic minority and European teenagers

Authors:

Sinead Brophy (s.brophy@swansea.ac.uk)
Annie Crowley (a.crowley@africaeducationaltrust.org)
Rupal Mistry (R.Mistry@comicrelief.com)
Rebecca Hill (r.a.hill@swanseac.uk)
Sopna Choudhury (s.m.choudhury@bham.ac.uk)
Non E Thomas (n.e.thomas@swansea.ac.uk)
Frances Rapport (f.l.rapport@swansea.ac.uk)

Version: 5 Date: 23 May 2011

Author's response to reviews:

Dear Editor

I have submitted another revision to make sure the authors are the same on the manuscript and on the submission system.

with good wishes and thanks

Sinead